Author's response to reviews

Title: In-hospital cerebrovascular complications following orthotopic liver transplantation: A retrospective study

Authors:

Li Ling (lingrirabbit@163.com)
Xiaoshun He (gdtrc@163.com)
Jinsheng Zeng (zengis@pub.guangzhou.gd.cn)
Zhijian Liang (lzj2004152000@yahoo.com.cn)

Version: 9 Date: 11 December 2008

Author's response to reviews:

2008-12-12

Dear Dr. Sabina Alam,

Thank you very much for accepting our manuscript entitled "In-hospital cerebrovascular complications following orthotopic liver transplantation: A retrospective study" (Manuscript ID 5770425052138950). We have now revised the manuscript following the editor's suggestions and the changes are listed below at a point-by-point style.

With best regards,

Sincerely yours,

Jinsheng Zeng, MD, PhD
Department of Neurology and Stroke Center,
First Affiliated Hospital, Sun Yat-Sen University,
No. 58 Zhongshan Road 2, Guangzhou, 510080, P.R. China.

Responses to the editor:

Table size must fit into an A4 portrait page. If you cannot re-orientate the table onto a single a4 portrait page, please move it to the additional files. Please ensure that you update the text to cite additional files and include a legend in the additional files section.

Response: We moved the table 2 to the additional files and updated the txt. A legend was included in the additional files section.

Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions as this is the final proofing stage for your manuscript, and you will not be able to make changes after acceptance.
Response: We checked the manuscript several times for typographical errors.